



# Article Developing Female Sustainable Entrepreneurial Intentions through an Entrepreneurial Mindset and Motives

Nadia Abdelhamid Abdelmegeed Abdelwahed \* D and Abdulrahman Alshaikhmubarak

Department of Management, Collage of Business, King Faisal University, Al-Ahsa 31982, Saudi Arabia; aalshaik@kfu.edu.sa

\* Correspondence: nabdelwahed@kfu.edu.sa; Tel.: +966-5-0719-6646

**Abstract:** In this study, we investigated the effect of entrepreneurial competencies (ECs) on entrepreneurial mindset (EM), entrepreneurial intention (EI), and entrepreneurial motives (EMTs) among female Saudi Arabian university students. We applied a quantitative approach and collected the data through a survey questionnaire by using convenience sampling to trace the respondents. Finally, the results of this study were based on 388 samples. By using a structural equation model (SEM), the findings demonstrate that ECs have positive and significant effects on EM, EI, and EMTs. In addition, there are positive and significant relationships between EM and EI as well as between EMTs and EI. Finally, EM and EMTs mediate the connection between ECs and EI. The findings of this study could support policymakers at different levels in developing their financial models, plans, and suggestions for entrepreneurial development. In addition, the findings should inspire well motivated people to engage in entrepreneurial activities and to aspire to become entrepreneurs. Finally, the originality and value of this research contribute to the existing literature, by demonstrating among Saudi Arabian female university students, the effects of ECs on EM, EI, and EMTs.

**Keywords:** entrepreneurial competencies (ECs); entrepreneurial mindset (EM); entrepreneurial motives (EMTs); entrepreneurial intention (EI); Saudi Arabian female university students; entrepreneurship

## 1. Introduction

Currently, entrepreneurship is well known for creating opportunities in economic development and changing global businesses and markets [1]. Entrepreneurship is important for the development of new ideas; however, it can also present risks [2,3]. Women's entrepreneurial skills have been shown to provide innovation and wealth development that can enable economic growth [4–6]. The empowerment of women is possible through entrepreneurship.

In Saudi Arabia, the government has devoted significant resources to foster an entrepreneurial culture, and entrepreneurship is a developing field. For instance, the Saudi Arabian government has created public and private institutions that provide services and support business owners when they start up their businesses [7,8]. These activities were also reflected in the 2019 GEM report which indicated that 76.3% of Saudi Arabia's adult population believed that the government offered appealing possibilities for launching a business [9]; moreover, according to the 2019 GEM report, roughly 33% of Saudi Arabia's population indicated their intentions to start businesses within the next three years. In addition, Saudi Arabia's total entrepreneurial activity (TEA) rate is relatively comparable to the GEM average. The 2019 GEM report expressed optimism about Saudi Arabia's business environment. Moreover, higher education institutes have made a very constructive contribution to nurturing entrepreneurship among women who are willing to start their own businesses and to become self-employed [2,5]. Therefore, there is a need to investigate EI among Saudi Arabia's female university students. EI refers to a person's conscious awareness and belief that they want to launch their own business in the future [8,10]. By



Citation: Abdelwahed, N.A.A.; Alshaikhmubarak, A. Developing Female Sustainable Entrepreneurial Intentions through an Entrepreneurial Mindset and Motives. *Sustainability* **2023**, *15*, 6210. https://doi.org/10.3390/su15076210

Academic Editor: Daniel Arias Aranda

Received: 8 March 2023 Revised: 31 March 2023 Accepted: 2 April 2023 Published: 4 April 2023



**Copyright:** © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). taking into consideration the significance and potential of this group of students, in this study, we examined these students' ECs on their EM, EI, and EMTs. Therefore, we aimed to answer the following research questions:

Q1: What are the effects of ECs on EM, EI, and EMTs among Saudi Arabia's female university students?

Q2: How do EM and EMTs mediate the association between ECs and EI among Saudi Arabia's female university students?

The findings of this study should assist policymakers to design policies that encourage female university students to fulfil their entrepreneurial activities and tasks. The findings should also provide university authorities with guidance to develop effective entrepreneurial education and technological advancement to divert female university students' intentions toward entrepreneurship and, thereby, eliminate unemployment. This paper is organized as follows: The introduction is provided in Section 1; Section 2 comprises the literature review and formulation of the hypotheses; Section 3 explains the study methods; Section 4 provides the data analysis; Section 5 presents the discussion and conclusions; and finally, Section 7 explains the limitations and implications of this study and makes recommendations for future research studies.

### 2. Literature Review and Formulation of the Hypotheses

### 2.1. Entrepreneurial Competencies (ECs)

ECs are crucial for business development and expansion. Government organizations and others frequently employ the idea of ECs to pursue economic growth and corporate success [11]. The value of ECs lies in their ability to predict the activity of an entrepreneur [12]. According to [13], factors such as ECs, competitive scope, and organizational capabilities can predict business performance. Entrepreneurship education programs help to develop ECs and the EI needed to start a business [14]. The authors of [15] suggested that ECs among female entrepreneurs, compared to their male counterparts, were more valuable for predicting small firm performance. The entrepreneurs' abilities to seize opportunities and their relational, innovative, human, and strategic qualities have direct and indirect impacts on SMEs' long-term performances [16]. Through entrepreneurial experiences, students can develop ECs [17]. In a developing context, the needs, opportunities, and start-up motivations of women entrepreneurs have direct, significant, and meaningful effects on small firm performance through the mediation of their ECs and motivation to learn [18]. The findings of [19] showed that ECs positively and significantly contributed to developing organizational capabilities, which, in turn, led to small firm performance. Organizational capabilities have a favorable influence on small firm performance and, to some extent, mediate the connection between ECs and small firm performance. In Australia and Malaysia, ECs are strong predictors of successful SMEs. Additionally, in Malaysia but not in Australia, ECs benefit from cultural orientations (both collectivism and tolerance for ambiguity) [20]. ECs are known as the characteristics of successful entrepreneurship. They are linked to the knowledge, attitude, and abilities that an entrepreneur must either possess or develop to achieve great results and to maximize business profits. These entrepreneurial skills are essential criteria for businesses to be successful [21]. Similarly, ECs mediate the association between entrepreneurial orientation and small firm performance. Furthermore, environmental dynamism reinforces the beneficial association between ECs and performance [22]. The findings of [23] acknowledged that ECs played vital roles in fostering SMEs' innovation and long-term success. Additionally, innovation partially moderated the relationship between ECs and manufacturing SMEs' sustainable performance. According to a study by [24], the performance of a spin-off during the "growth" phase was positively impacted by the entrepreneurial capabilities of the original team that were strengthened during the "creation" phase. Through the development of a team's entrepreneurial capabilities, the original team's networks had secondary impacts on the success of a spin-off. The dimensions of the entrepreneurship attitude orientation (EAO) measure, except for personal control, were reported to be substantially and favorably affected by the gamified simulation

experience among students from various high schools in the region of Granada (Spain). It is highly desirable to include gamified simulation activities in high school courses to, among other beneficial impacts, lower the psychological barriers relating to self-esteem or an absence of innovation that may prevent individuals from becoming entrepreneurs [25]. The authors of [26] advocated that ECs played beneficial and significant roles in the management of a business's control system. Corporate strategy (cost leadership and differentiation strategy) has a significant influence on business performance, whereas ECs link cultural planning, cybernetics, rewards and pay, administrative control, and business performance. In [27], the authors stated that the ability to identify exploitable opportunities and the competency to recognize opportunities are two different concepts. Students who are skilled at identifying opportunities are motivated to start their own businesses. However, they may need either more skills or opportunities to make this happen. It has been reported that entrepreneurial knowledge, entrepreneurial awareness, and absorptive capacity are significant predictors of competencies in recognizing opportunities.

### 2.2. Entrepreneurial Mindset (EM)

In recent years, EM has become more prevalent in discussions of entrepreneurship, as researchers examine its origins, mechanisms, and manifestations [2]. Defined as the ability to perceive, act, and mobilize in ambiguous situations [28], EM is an important factor for understanding how entrepreneurs approach their work [29]. Through an entrepreneurialdirected approach to education, students gain a better understanding of entrepreneurship, which helps them to develop the behaviors and abilities necessary for academic success [30]. Teachers use design thinking methodologies to guide students' projects and to assess students' EM development [31].

EM is affected by expectations, entrepreneurial identity, as well as self-efficacy [32]. As a result, entrepreneurship education in higher education institutions has increasingly aimed to foster students' EM [33]. To effectively support students' EM, entrepreneurship education and self-efficacy must be mediated [34]. While entrepreneurial education also affects EM, attitude, and self-efficacy, it is important to note that entrepreneurial self-efficacy specifically encourages an entrepreneurial mindset.

### 2.3. Entrepreneurial Motives (EMTs)

Research has shown a positive correlation between narcissism and entrepreneurial goals among MBA students. Additionally, a subgroup analysis of high-EI business undergraduate and MBA students, specifically early-stage fledgling entrepreneurs, revealed a range of motivations for participation in the startup process [35]. The findings of [36] confirmed that, for 80% of informal entrepreneurs, starting a business was motivated by necessity as well as opportunity. However, as their initiatives gained traction, their motivations shifted toward opportunity drivers.

Employing a PLS-SEM analysis, ref. [37] found that entrepreneurial mindset traits (i.e., EMTs) influenced attitude toward a meal-sharing economy and the desire to participate in it. Participants' attitudes toward a sharing economy were affected by factors such as independence, social interactions, and gratification from hosting.

According to [38], while economic survival is the primary motivation for Mexican business people, Canadian and American business people value intrinsic pleasures more than perceptions of success. The research by [39] revealed entrepreneurial obstacles that are common to all emerging economies, despite situational differences. Motivations are also linked to corporate resources, behaviors, and performance [40]. In addition, there are significant variations in typical salary levels and the number of years in business.

### 2.4. Entrepreneurial Intention (EI)

The achievement of entrepreneurial goals is positively correlated with effective family– work enrichment through the mediating role of entrepreneurial self-efficacy. This relationship is particularly strong for individuals who prefer less work–home segmentation [41]. While prison entrepreneurship programs have been shown to have no direct impact on prisoners' EIs, they have been found to enhance prisoners' self-efficacy and entrepreneurial resilience, which in turn, promote the development of EI [42]. However, previous studies have suggested that entrepreneurial resilience does not affect the EIs of prisoners who have no prior experience with entrepreneurship training [43].

Education, gender, diversity education, and owner's minority status have significant impacts on digital innovation [43]. Attitude is a strong predictor of decision making among Italian university students [44], and there are differences between men and women in their perceptions of cognitive, economic, and social recognition [45]. Perceived self-efficacy and perceived desirability are significant factors that enable EI among business students in Pakistani universities, and entrepreneurial knowledge moderates these associations [46]. Using the theory of planned behavior (TPB), ref. [47] suggested the positive and significant effect of factors such as personnel attitude, EMTs, subjective norms, and perceived behavioral control on EI among female students. Likewise, with the help of the TBP, a longitudinal study by [48] confirmed a positive and significant correlation between attitude toward entrepreneurship and EI. By employing the entrepreneurial event model (EEM), scholars such as [49,50] claimed a predictive and positive effect of self-efficacy, perceived desirability, and perceived feasibility on EI among the students of Pakistan.

While self-efficacy can play an influential role in resolving conflicts, there is a positive correlation between perceived impediments and entrepreneurial exit intentions [51]. During the COVID-19 pandemic, the desire to start a business has been essential in mediating job uncertainty and green EI, with entrepreneurial passion as a mediator [52]. Demographic, social, and environmental factors significantly influence the EI of engineering undergraduates, while age and occupation do not indicate a person's intentions to work for themselves [53]. The antecedents of future entrepreneurs, such as role models and innovation, significantly influence digital entrepreneurial decisions, and digital competence increases the propensity to launch digital businesses [54,55].

Personality traits, such as conscientiousness, have a positive impact on EI, while neuroticism has a negative effect [55]. Entrepreneurial education has been found to modulate the impacts of demand for achievement, locus of control, and innovation on EI to some extent [56]. The factors that affect EI vary between genders; male students and students whose parents either own businesses or have retired have higher EI, while female students are affected by subjective norms, attitude, and perceived behavioral control [33,57]. Social pressures have been reported to affect the intention of female Italian business students to become entrepreneurs [58].

Several studies have investigated EI directly, using various factors such as entrepreneurial motivation, need for achievement, entrepreneurship attitude, subjective norms, innovation, personality traits, entrepreneurial orientation, ECs, EM, entrepreneurial education, EMTs, organizational skills, subjective beliefs, and career opportunities [27,59–61]. In Saudi Arabia, EI has been examined through university students' ECs and EM, but the roles played by EM and EMTs in mediating the relationship between ECs and EI still need further exploration [2].

To address these gaps in the existing literature, in this study, we developed a model (see Figure 1) to confirm the effect of ECs on EM, EI, and EMTs among female Saudi Arabian university students.



5 of 20



Figure 1. Conceptual model of this study (source, conceptualized by the researchers).

### 2.5. Entrepreneurial Competencies (ECs) and Entrepreneurial Mindset (EM)

In order to achieve success in any venture, networking is a crucial factor. Entrepreneurial and managerial core competencies are also essential, with ECs being slightly more significant. Specifically, the number of years of entrepreneurial experience, entrepreneurial family background, and family situation are significant factors [62]. Technological intensity and entrepreneurial orientation are the primary drivers of managerial mindset intensity on a global scale [63]. The findings from [64] shed light on how information literacy activities designed to strengthen EM changed students' perceptions. Similarly, Greek science and economic graduates' beliefs were reduced through entrepreneurship education. An entrepreneur's emergence is influenced by various factors such as personal skills, self-assurance, organizational skills, and the adoption of an EM. However, a potential misunderstanding of entrepreneurial principles can lower EI [65]. In the UAE, entrepreneurship is the preferred initial professional path for young people, and environmental and individual factors impact men's and women's EM [66]. However, ref. [67] found differences in partnership accounting and the attainment of social and functional ECs between a project class and a control class. In emerging economies such as Ghana, fostering an international mindset promotes entrepreneurial behaviors that drive SMEs' performances [68].

According to [69], teachers who set an example for students see a significant rise in students' ECs and perceived behavioral controls throughout the semester; however, their attitude toward intention decreases. ECs and EM are significant predictors of SMEs' performances in Nigeria's Benue State [70]. Positive EI development has been linked to female students' ECs related to EM. Among university students who had high EM, there was a high correlation between ECs and EM, but only a moderate correlation between each dimension of self-entrepreneurial competencies [60]. Furthermore, EM mediates the association between ECs and EI [2]. Based on the positive associations found, in this study, we formulated the following hypothesis:

**H1.** ECs have a positive and significant relationship with EM.

### 2.6. Entrepreneurial Competencies (ECs) and Entrepreneurial Intention (EI)

Aspiring entrepreneurs who excel at identifying opportunities may require more concrete ideas or exploitable opportunities to pursue. According to the research by [27,69], having an international mindset can enhance the positive effects of innovative entrepreneurial and risk-taking behaviors. Students who are mentored by teachers who serve as role models tend to have higher levels of perceived behavioral control and entrepreneurial confidence, while those who lack such guidance exhibit poorer attitudes toward entrepreneurship.

Entrepreneurship education programs have been shown to positively influence university students' ECs and self-employment intentions, as demonstrated in [14,71]. Competency clusters play significant roles in the development of EI, and there is evidence of a cognitive bias toward overestimation, with certain competency clusters having indirect effects through higher level clusters [72].

In Bali, Indonesia, personality traits significantly influence ECs and EI, with a bidirectional relationship between the two, as reported in [73]. However, in Iran, opportunity alertness is less prevalent, and the most common ECs among adults are self-efficacy, risk taking, and role modeling, although ECs do not translate to increased risk propensity in terms of EI. Instead, Iranians' EIs are primarily influenced by self-efficacy [74].

In contrast to these findings, ref. [75] found no significant connection between entrepreneurial action and recurring entrepreneurial intention behaviors. To investigate whether such associations are positive or negative, we formulated the following hypothesis:

### **H2.** ECs have a positive and significant relationship with EI.

### 2.7. Entrepreneurial Competencies (ECs) and Entrepreneurial Motives (EMTs)

ECs develop students' skills to become successful entrepreneurs [76]. Using multiple regression analysis, the findings by [77] confirmed the relationship between educational attainment, ECs, culture, and EMTs. The findings by [15] underlined the significant effects of ECs on small firm performance and growth. In the Nigerian context, ECs and EO directly influenced SMEs' performances [78]. In Malaysian SMEs, ECs were strong predictors of successful businesses. Compared to more benign and stable contexts, hostile and dynamic environments have shown a stronger correlation between ECs and a successful business [79]. The findings by [80] demonstrated that economic vulnerability substantially damaged commitment, opportunity recognition, organization, and strategic skills. Conversely, the findings showed that, while economic vulnerability had a considerable positive impact on competency, it had a marginally beneficial impact on conceptual competency. Similarly, entrepreneurial learning has direct and indirect effects on ECs and venture growth. Entrepreneurs have used venture expansion as a learning tool to become competent [81]. Factors such as EM, ECs, self-efficacy, and the TBP factors (attitudes, subjective norms, and perceived behavioral control) have been shown to positively affect venture creation and EI [82,83].

The entrepreneurial approach has a clear and positive association with entrepreneurs' strategic and commitment competencies [84]. In Sri Lanka, owners'/managers' background traits directly affect ECs. Based on the positive relationships in the literature, we formulated the following hypothesis:

### **H3.** *ECs have a positive and significant relationship with EMTs.*

#### 2.8. Entrepreneurial Mindset (EM) and Entrepreneurial Intention (EI)

An EM contributes significantly to a student who chooses entrepreneurship as their future employment [66]. In Ukraine's universities, students receiving entrepreneurship-specific education demonstrated a more intense entrepreneurial attitude when they had accumulated more entrepreneurial alertness assets [85]. Entrepreneurship is the first choice for the UAE's young professional people. However, most of them have not yet enrolled in any high school or college formal entrepreneurship courses [66]. According to [86], the determinants of entrepreneurship, professionalism, and leadership are self-efficacy, EI,

entrepreneurial and job outcome expectations, socioeconomic level, and an open personality. Through its ability to innovate, a creative mindset enables successful entrepreneurship. Entrepreneurs who have a growth mentality show and fuel their successes either directly or through their capacity for innovation [87]. The findings by [88] showed that selfefficacy was successful in accelerating university students' EIs. In part, an entrepreneurial culture impacted on entrepreneurship education and EI. In addition, there was a strong association between students' EM and their entrepreneurial culture and schooling. Contrary to predictions, this study's findings did not show that there was a significant relationship between students' EIs and entrepreneurial instruction. According to [89], entrepreneurial education activities influenced entrepreneurial behaviors, and behavioral EM mediated this link. Likewise, entrepreneurial self-efficacy has been reported to be a substantial predictor of identity and education [32]. In the same domain, applying the TPB theory, ref. [90] found a significant positive connection between EI and subjective norms, attitude toward entrepreneurship, and perceived behavioral control among university students. The EM factor was a direct and indirect enabler of a predictor of EI. Similarly, in China, the TBP theory also has a valid and predictive effect on EI, as EM boosts EI [91]. Among science and technology students, entrepreneurship education, the TBP factors (attitude, subjective norms, and perceived behavioral control), and EM positively affect EI [92,93]. In Nigeria, contextual and individual factors have a significant relationship with EI, and self-efficacy partially mediates this relationship [94]. Therefore, we formulated the following hypothesis:

### **H4.** EM has a positive and significant relationship with EI.

### 2.9. Entrepreneurial Motives (EMTs) and Entrepreneurial Intention (EI)

While sympathy and empathy are predecessors of other-oriented motivations, such as charity and social justice, entrepreneurial enthusiasm and impatience lead to self-oriented objectives [95]. An entrepreneurial business is a family's primary source of income [96]. Similarly, ref. [97] tested EI toward perceptions of motives and barriers regarding entrepreneurship. The study's findings pointed to several essential variations between the two sample groups and meaningful connections among the explanatory factors, goals, motives, and impediments. Needs and opportunities both influenced entrepreneurs' informed decisions to launch their businesses. As their initiatives gained more traction, there was also a visibly discernible change in their motivations away from necessity and toward opportunity drivers [36]. The findings by [98] showed that financial motivations were unrelated to either perceived political or governmental, or commercial prospects. However, the anticipation of favorable government policies and marketing possibilities were consistently linked to the drive for recognition. Based on the existing association of EMTs with EI, we formulated the following hypothesis:

### **H5.** *EMTs have a positive and significant relationship with EI.*

### 2.10. Entrepreneurial Mindset (EM) and Entrepreneurial Motives (EMTs) as Mediators

Entrepreneurship education is crucial to provide students with the knowledge and skills necessary for successful careers in entrepreneurship. By positively influencing entrepreneurial EM and practices, entrepreneurial knowledge plays a key role in mediating the effects of entrepreneurial education and preparation [99]. EM has also been found to accelerate university students' EIs [88]. Entrepreneurial culture is another important factor that influences entrepreneurship education and EI, with a strong link between students' EM and entrepreneurial culture and education [61].

Research has shown that entrepreneurship education increases students' entrepreneurial aspirations by fostering the development of their EM, while also having favorable and significant impacts on entrepreneurial alertness and EM [100]. Furthermore, EM plays a significant mediating role in the relationship between entrepreneurial awareness and education [101]. Creative thinking also has a positive impact on the association between EM and corporate entrepreneurship [102].

In Indonesia, entrepreneurship education is a predictor of students' EIs and EM, with a robust connection between EM and students' EIs [103]. EM also plays a meaningful role in driving entrepreneurial activity toward EI [104]. Entrepreneurship education not only facilitates individual entrepreneurial orientation and EM, but also has a positive association with EI. Moreover, EM significantly mediates the relationship between individual entrepreneurial orientation and EI, as well as between entrepreneurship education and EI [105].

Factors such as risk taking, need for achievement, environmental support, and locus of control are positively and significantly associated with each other, with students' personalities and EM mediating these associations [59]. Social entrepreneurs have also found that other-oriented motives are significantly related to their work, while self-oriented motives are not. Furthermore, the level of perceived work competence mediates the relationship between enthusiasm and job stress, indicating that high levels of work competence can benefit social entrepreneurs' enthusiasm and can strengthen the association between enthusiasm and other-centered goals [106].

Overall, EM and EMTs play significant mediating roles in the relationship between ECs and EI, as evidenced by previous research [61,64,88,103,104]. However, there is a lack of integrated research on the role of EM and EMTs in driving EI [99–101,105,106]. Based on this deficiency, we formulated the following hypotheses:

H6. EM mediates the relationship between ECs and EI.

**H7.** *EMTs mediate the relationship between ECs and EI.* 

### 3. Methods

3.1. Survey Strategy and Respondents

In this study, we used the quantitative technique and numerous quantitative data to explore the link [107]. This approach is excellent and essential to gathering data in a systematic manner [108]. Generally, when conducting either a quantitative evaluation or empirical inquiries, this approach is more commonplace and consistently gives a quantitatively correct representation of society [1,109,110]. In entrepreneurship and, more particularly, in the context of EI, several scholars such as [2,64,67,69,73,89] have applied the quantitate technique to assess EI.

This study was conducted between August 2022 and January 2023. The study's respondents were female students enrolled at different Saudi Arabian public and private universities because they were regarded as potential entrepreneurs and about to make decisions regarding their future professional lives [111,112]. The context of this study is that Saudi Arabia is a developing nation with a rapidly expanding economy [113]. According to [114], a country's unfavorable environmental conditions can make it difficult for women to start businesses. Compared to their Western counterparts, Middle Eastern women face various difficulties, such as social discrimination problems stemming from gender stereotypes, traditional issues, cultural concerns, and business system regime problems [5,115,116]. In Western countries, where laws forbid discrimination, men and women are treated equally. However, there is some gender-based stereotyping in Middle Eastern countries and women are not given the same respect as men [5]. The Saudi government has recently established several programs to encourage female engagement in entrepreneurship. This is despite traditional Saudi conventions representing obstacles for women who want to engage in such activities [6]. Due to women's challenges, there is a need to increase the number of women who can benefit from these initiatives [4]. According to the Saudi Vision 2030, the unemployment rate must drop from 12.9% to 7% [117]. The Saudi Arabian government has committed significant funding to universities to foster their students' entrepreneurial spirits and to inspire them to participate in entrepreneurship [118]. Keeping this in mind, we examined Saudi Arabian university students' ECs, EM, EMTs, and EI.

### 3.2. Instrument and Reliability

The items on the scales had been previously validated; however, to ensure their validation in the context of Saudi Arabia, we conducted a pilot study to ensure the reliability and validity of the questionnaire [119] which we derived from the literature. In terms of its reliability, we performed the most used method for estimating reliability, namely internal consistency coefficients through using Cronbach's alpha [120]. Consequently, the overall reliability was greater than 0.70 with individual factors and satisfactory scores greater than 0.70 [121]. The study's validity is the degree to which the rules of scientific research methodology are adhered to when producing the research results. This is a prerequisite for all academic research studies [122]. The validity of a study tool is determined by how well it measures what it is intended to measure [123]. Content validity research can provide information on each item's representativeness and clarity along with a preliminary review of its factorial validity. In this regard, field experts/university professors offered constructive feedback about this study's objectives and the quality of the measure of the data collected by the questionnaire [124]. The field experts provided concrete suggestions or recommendations to improve the content of the survey instrument [125]. Therefore, the questionnaire was sent to university professors who were in the entrepreneurship field and were familiar with the research methodology to ensure its content, format, and design. The phrasing of each question was straightforward and understandable to the chosen sample, and therefore, there were no issues. Consequently, after making some minor modifications, we distributed a reliable and valid questionnaire to collect large-scale data.

### 3.3. Data Collection and Sample Size

We used the questionnaire, both online and offline, to collect data from the targeted female university students enrolled at different Saudi Arabian public and private universities. The study's target population was all female students pursuing their undergraduate and graduate degrees. Because most previous EI research studies had included samples of college students, we concentrated on university-based female students since they would be making decisions about their future professional careers and were prospective entrepreneurs. We applied convenience sampling since it was the optimum method for online and offline surveys [126]. We visited the Saudi Arabian universities and sent e-mails to a select group of students and WhatsApp groups along with links to the questionnaire. We attached a covering page describing the study's objectives and the students' voluntary participation. We also assured the respondents about the privacy and confidentiality of their responses.

We distributed 600 questionnaires to the students and received back 388 samples which represented a 64% response rate. We found no cases with less than 5% and no outlier cases. Therefore, we based the final analysis on 388 valid cases.

We used G\*Power (version 3) to calculate the required sample size, which is a good freeware program that determines statistical power for sample size analysis in behavioral research for the most common statistical tests [127,128]. We used all four variables to guarantee a sufficient sample size. Therefore, G\*Power stated that 100 samples were required to conduct the SEM analysis, and to fulfil the AMOS software's requirement.

Having collected the data, we applied a *t*-test to identify the mean difference between both sampling modes. We found a significant difference in mean scores at <0.005 \*\* or a two-tailed significance level (see Table 1). In this way, we accomplished the existence of probably a statistical difference between the online and offline samples [129] and in doing so, ensured no assumption either of response bias or common variance.

Offline	Online	df	Sig.(2-Tailed)	Mean Difference	95% Confidence Interval of the Difference		
						Upper	
230 (59.28%)	158 (40.72%)	321	< 0.005 **	5.007	2.624	4.725	
The explanation of **. $p < 0.001$							

**Table 1.** Mean difference in offline and online samples.

· ·

### 3.4. Measures

Entrepreneurial Competencies (ECs)

ECs underline an individual's skills, capabilities, or proficiencies utilized to develop entrepreneurial and financial plans, enterprise management models, resources, and novel ideas to succeed in a business [130,131]. We used six items adopted from [10,130,131] to measure ECs. A sample item is: "I can develop an entrepreneurial proposal".

Entrepreneurial Mindset (EM)

An EM underlines an individual's capacities to perceive, act, and mobilize in ambiguous situations [28]. It shows an individual's interest, motivation, and inspiration to be involved in an entrepreneurial activity, a new business, and to create a new venture [132–134]. We used four items, adapted from [132–135], to measure EM. A sample item is: "I am interested in entrepreneurial activities".

Entrepreneurial Motives (EMTs)

EMTs underline an individual's drive or motivation toward entrepreneurship, where they want to be self-employed, serve, and host a society; they want to be richer and to develop their entrepreneurial careers, which attracts them, and entrepreneurship runs in their blood [136]. We used three items, adopted form [136], to measure EMTs. A sample item is: "A career as an entrepreneur is attractive for me".

Entrepreneurial Intention (EI)

EI refers to an individual's conscious state of mind toward performing an entrepreneurial behavior, i.e., becoming an entrepreneur and launching a new business in the future [8,10]. We used six items, adopted from [2] as utilized by [130,137–139], to measure EI. A sample item is: "I have decided on a new creation in the future".

We evaluated all the items by using a five-point Likert scale based on the following options: strongly agree = 1, agree = 2, neutral = 3, disagree = 4, and strongly disagree = 5.

### 4. Analysis

### 4.1. Demographic Indicators

The researchers observed a few demographic indicators such as age, university, and discipline, to understand the dynamics of the study population. We collected the data from a total of 388 respondents; 71.40% (n = 277) of the students who contributed to this study were from 21 to 30 years old, 23.45% (n = 91) were aged below 20 years, and 5.15% (n = 20) were 31 years and above. Related to the universities, most respondents (68.56% or n = 266) were from public universities, and 31.44% (n = 122) were from private universities. Regarding disciplines, most students (24.74% or n = 96) were from engineering, 24.23% (n = 94) were from management, 22.16% (n = 86) were from business, 19.85% (n = 77) were from information technology, and only 9.02% (n = 35) were from other disciplines (see Table 2).

Table 2. Demographic indicators.

Indicator	Characteristics	Samples	Percentage		
Age	<20 years	91	23.45		
(years)	21–30	277	71.40		
	31 and above	20	5.15		
	Total	388	100.0		

				_
Indicator	Characteristics	Samples	Percentage	
University	Public	266	68.56	
-	Private	122	31.44	
	Total	388	100.0	
Discipline	Engineering	96	24.74	
-	Business	86	22.16	
	Management	94	24.23	
	Information technology	77	19.85	
	Others	35	9.02	
	Total	388	100.0	

Table 2. Cont.

### 4.2. Measurement of Model Assessment

We conducted factor loading to determine the correlation coefficients between the constructs. Most items had loading values larger than 0.70 or were within a range between 0.701 (emts7) and 0.890 (em1). However, certain items, including ecs4, emts2, emts8, emts11, and ei4, were disqualified because either their loading weights needed to be increased or their required values needed to be met [121]. Likewise, we noted that the composite reliability (CR) values ranged from 0.761 (EMTS) to 0.800 (EM) or were higher than the suggested values (0.70) (see Table 3). We computed the average variance extracted (AVE) values to identify and to evaluate the elements further. Accordingly, all the AVE construct values were between 0.781 and 0.826, this is more than 0.50 and is regarded to be excellent [121]. Finally, we used Cronbach's alpha reliability to evaluate the items' internal consistency. We noted that Cronbach's alpha ranged from 0.820 (for EMTs) to 0.873 (EI). This shows a satisfactory dependability for all factors (>0.70) (see Table 3).

Construct	Code	Loadings	CR	AVE	α
Entrepreneurial competencies	ecs1	0.882	0.798	0.792	0.869
(ECs)	ecs3	0.866			
	ecs2	0.857			
	ecs5	0.838			
Entrepreneurial mindset	em1	0.890	0.800	0.781	0.882
(EM)	em2	0.867			
	em3	0.833			
	em4	0.829			
Entrepreneurial motives	emts1	0.878	0.761	0.799	0.820
(EMTs)	emts3	0.859			
	emts5	0.828			
	emts4	0.810			
	emts9	0.798			
	emts13	0.765			
	emts10	0.749			
	emts12	0.730			
	emts6	0.719			
	emts7	0.701			
Entrepreneurial intention	ei1	0.872	0.789	0.826	0.873
(EI)	ei2	0.862			
	ei3	0.844			
	ei6	0.820			
	ei5	0.808			

 Table 3. Measurement model.

Note: CR, square of the summation of the factor loadings; AVE, summation of the square of the factor loadings;  $\alpha$ , Cronbach's alpha.

It is also essential to have a well-fitting measurement model before studying causal pathways in a structural model [140]. If good-fitting models are consistent with the data,

these can be re-specified. Before assessing the hypotheses, we evaluated the model fit to identify its ability to replicate the data (i.e., usually the variance-covariance matrix). As shown in Table 4 and Figure 2, the chi-square/df values are 2.270 and, thereby, ensure the model's initial compatibility with the data. In addition, within the acceptable scores suggested by researchers such as [140–142], we discovered other model fit indicators such as CFI, GFI, AGFI, NFI, and RMSEA (Table 4).



**Figure 2.** Structural equation model (source, authors' own estimation). Note: CR, critical ratio; \*\*\* p < 0.05; ECs, entrepreneurial competencies; EM, entrepreneurial mindset; EMTs, entrepreneurial motives; EI, entrepreneurial intention; CMIN,  $\chi^2$ /chi-square/df; df, degrees of freedom; GFI, goodness-of-fit index; AGFI, adjusted goodness-of-fit index; NFI, normed fit index; CFI, comparative fit index; RMSEA, root mean square error of approximation.

Goodness of Fit Indices	Achieved Value	Acceptance Level
Chi-square/df	2.270	<5.0
ĊFI	0.900	>0.90
GFI	0.932	>0.90
AGFI	0.918	>0.85
NFI	0.922	>0.90
RMSEA	0.037	<0.08

Table 4. Model fit indices and their acceptable thresholds.

Note: CMIN,  $\chi^2$ /chi-square/df; df, degrees of freedom; GFI, goodness-of-fit index; AGFI, adjusted goodness-of-fit index; NFI, normed fit index; CFI, comparative fit index; RMSEA, root mean square error of approximation.

#### 4.3. Structural Model Assessment

We applied a SEM path analysis to assess the developed hypotheses. The analysis shows that EMTs have a positive and significant effect on EI (H5 = SE = 0.029, CR = 5.182, \*\*\* p < 0.01). Turning to the direct paths, as shown in Table 5 and Figure 2, these confirm that ECs have a positive and significant relationship with EM (H1 = SE = 0.024, CR = 6.091, \*\*\* p < 0.01). Therefore, hypothesis H1 is accepted. Likewise, ECs have a positive and significant relationship with EI (H2 = SE = 0.032, CR = 6.009, \*\*\* p < 0.01). Therefore, hypothesis H2 is accepted. By showing the positive and significant paths (H3 = SE = 0.026, CR = 7.391, \*\*\* p < 0.01), the analysis also demonstrates the association between ECs and EMTs. Therefore, hypothesis H3 is accepted. The analysis also shows that EM has a positive and significant predictive power on EI, (H4 = SE = 0.026, CR = 5.690, \*\*\* p < 0.01). Therefore, hypothesis H4 is accepted. Finally, EMTs have a positive and significant effect on EI (H5 = SE = 0.029, CR = 5.182, \*\*\* p < 0.01) (see Table 5 and Figure 2). Therefore, hypothesis H5 is accepted.

Table 5. SEM estimations (direct paths).

H.No	Independent Variables	Path	Dependent Variables	Estimate	SE	CR	р	Decision
H1	ECs	$\rightarrow$	EM	0.288	0.024	6.091	***	Accepted
H2	ECs	$\rightarrow$	EI	0.241	0.032	6.009	***	Accepted
H3	ECs	$\rightarrow$	EMTs	0.249	0.026	7.391	***	Accepted
H4	EM	$\rightarrow$	EI	0.032	0.026	5.690	***	Accepted
H5	EMTs	$\rightarrow$	EI	0.239	0.029	5.182	***	Accepted

Note: SE, standard error; CR, critical ratio; \*\*\* p < 0.05; ECs, entrepreneurial competencies; EM, entrepreneurial mindset; EMTs, entrepreneurial motives; EI, entrepreneurial intention

Moreover, in relation to the indirect paths, the analysis shows that EM has an indirect effect on EMTs in developing the association between ECs and EI (H6 = SE = 0.020, CR = 5.562, \*\*\* p < 0.01 and H7 = SE = 0.023, CR = 6.172, \*\*\* p < 0.01 (see Table 6 and Figure 2). Therefore, hypotheses H6 and H7 are accepted.

Table 6. SEM estimations (indirect paths).

H.No	Independent Variables	Path	Mediator	Path	Dependent Variables	Estimate	SE	CR	р	Decision
H6	ECs	$\rightarrow$	EM	$\rightarrow$	EI	0.211	0.020	5.562	***	Accepted
H7	ECs	$\rightarrow$	EE	$\rightarrow$	MTE	0.230	0.023	6.172	***	Accepted

Note: SE, standard error; CR, critical ratio; \*\*\* p < 0.05; ECs = Entrepreneurial competencies; EM = Entrepreneurial mindset; EMTs = Entrepreneurial motives; EI = Entrepreneurial intention.

#### 5. Discussion and Conclusions

In this study, we explored the relationships of ECs with EI among Saudi Arabian female university students. We also examined the mediating roles of EM and EMTs. The findings show that ECs have positive and significant effects on EM, EI, and EMTs. These

positive findings are consistent with those of previous studies such as [62–64,69,71,73,79,80]. These findings demonstrate that female students can create business plans and can realize enterprise management models. They have access to business resources. At various phases, they can carry out the financial strategy for entrepreneurship development. They can apply creative concepts to marketing. They have an interest in business ventures. They aim to start their own businesses. They will launch a new business if they have the opportunity and the resources. They will choose a brand new creation from among the available professional options. They want more profound integration with the host culture. Their entrepreneurship aids their integration. The benefits of being an entrepreneur outweigh the drawbacks. They find an entrepreneurial profession appealing. Being an entrepreneur is incredibly satisfying. They prefer having their own businesses to being in stable employment. When people work for themselves, they can earn a good income. Rather than managing an existing company, they prefer to start their own businesses. They desire independence. They like the freedom that working for themselves provides. Their blood is infused with entrepreneurship. They desire financial success.

This study's findings demonstrate that EM has a positive and significant effect on EI. These findings are consistent with the findings by [87–89,94]. EM significantly contributes to students choosing entrepreneurship as their future employment and careers. The students can create an entrepreneurial proposal while also developing positive EI. They also recognize the significance of enterprise management models in improving EI. They can acquire entrepreneurial resources and can set a financial plan for various stages of entrepreneurial development and, more particularly, for EI. They intend to apply their novel ideas to the advancement of marketing.

In addition, this study's findings demonstrate that EMTs have a positive and significant effect on EI. Likewise, these findings are consistent with those of previous studies such as [95–98]. Saudi Arabian female university students are interested in entrepreneurial activities. They want to be business owners based on the assumption that they will start a new business if they have enough opportunities and resources. They will choose a new business from among the various career options and because of such factors, they may develop an interest in entrepreneurship. The mediating effects show that EM and EMTs are the significant factors which develop the association between ECs and EI. The indirect relationship between these factors suggests that they have decided to create something new in the future. Five years following graduation, they will have started new businesses since they have prepared and made plans to be business owners. They are confident in their ability to launch a successful new business. EM and EMTs reinforce all these intentions and improve the connection between ECs and EI.

In summary, this study's overall findings show that ECs, EM, and EMTs reinforce EI. The findings also confirm the positive predictive power of EM and EMTs on EI. This significantly enhances EI among female university students. Moreover, EM and EMTs indirectly develop the relationship between ECs and EI. These connections make possible the enhancement of EI through entrepreneurial education and university education in IT, management, business, and engineering.

### 6. Study Implications

### 6.1. Managerial Implications

This findings of this study have several implications for policymakers, educators, and business leaders. First, efforts should be made to develop and implement entrepreneurship education programs that are targeted specifically at female university students and are focused on enhancing their ECs, EM, and EMTs. Second, universities should establish mentoring and coaching programs to provide female students with guidance, support, and role models as they explore entrepreneurship. Finally, policymakers should implement policies and initiatives that address the social and cultural barriers that limit Saudi Arabian women's participation in entrepreneurship. This can be done through providing access to financing, improving the legal and regulatory environment, and promoting gender equality.

### 6.2. Theoretical Implications

This study makes several important contributions to the literature on entrepreneurship and EI among Saudi Arabian female university students. First, it highlights the critical role played by ECs in shaping EI. This underpins the importance of enhancing ECs among Saudi Arabian female university students as key drivers of entrepreneurship. Second, the findings demonstrate the crucial mediating role played by EM and EMTs in linking ECs to EI. These findings demonstrate that, in addition to developing ECs, efforts should be made to promote greater EM and EMTs among female university students to increase the likelihood of their pursuing entrepreneurship. Finally, this study's findings shed light on the unique context of female entrepreneurship in Saudi Arabia where gender norms and social barriers have often limited women's participation in the labor force and entrepreneurship. This study's findings can inform policies and programs aimed at increasing women's participation in entrepreneurship and in promoting gender equality.

### 7. Limitations and Future Research Directions

One limitation of this study was that it relied on cross-sectional data. This limited our ability to draw causal conclusions about the relationships among ECs, EM, EMTs, and EI.

Consequently, we recommend that future research studies should use either longitudinal designs or experimental methods to establish the causal relationships among these variables. In addition, this study only focused on Saudi Arabian female university students, and therefore, the findings cannot be generalized to other contexts or populations. Consequently, the researchers recommend that future research studies should explore the relationships between these variables in other populations such as male university students, working professionals, or entrepreneurs. Finally, this study did not examine the influence of contextual factors, such as the social and cultural environment, on the relationships between these variables. Consequently, the researchers recommend that future research studies should explore how these contextual factors shape the relationships among ECs, EM, EMTs, and EI among Saudi Arabian female university students and other contexts.

Author Contributions: Conceptualization, N.A.A.A. and A.A.; Data curation, A.A.; Formal analysis, N.A.A.A.; Funding acquisition, N.A.A.A.; Investigation, N.A.A.A.; Methodology, N.A.A.A. and A.A.; Project administration, N.A.A.A. and A.A.; Resources, A.A.; Software, N.A.A.A.; Supervision, N.A.A.A.; Validation, N.A.A.A. and A.A.; Visualization, N.A.A.A.; Writing—original draft, N.A.A.A.; Writing—review and editing, N.A.A.A. and A.A. All authors have read and agreed to the published version of the manuscript.

**Funding:** This work was supported by the Deanship of Scientific Research, Vice Presidency for Graduate Studies and Scientific Research, King Faisal University, Saudi Arabia (Project No. GRANT 3160).

**Institutional Review Board Statement:** This study was conducted according to the guidelines of the Declaration of Helsinki and approved by the deanship of scientific research ethical committee, King Faisal University (project number: GRANT 3160).

Informed Consent Statement: Informed consent was obtained from all subjects involved in the study.

**Data Availability Statement:** Data are available upon request from researchers who meet the eligibility criteria. Kindly contact the first author privately through e-mail.

Conflicts of Interest: The authors declare no conflict of interest.

### References

- 1. Soomro, B.A.; Memon, M.; Shah, N. Attitudes towards entrepreneurship among the students of Thailand: An entrepreneurial attitude orientation approach. *Educ.* + *Train.* **2020**, *63*, 239–255. [CrossRef]
- Chang, A.; Chang, D.F.; Chen, T.L. Detecting Female Students Transforming Entrepreneurial Competency, Mindset, and Intention into Sustainable Entrepreneurship. Sustainability 2022, 14, 12970. [CrossRef]
- Pidduck, R.J.; Clark, D.R.; Lumpkin, G.T. Entrepreneurial mindset: Dispositional beliefs, opportunity beliefs, and entrepreneurial behavior. J. Small Bus. Manag. 2023, 61, 45–79. [CrossRef]

- 4. Ali, I.; Ali, M.; Badghish, S. Symmetric and asymmetric modeling of entrepreneurial ecosystem in developing entrepreneurial intentions among female university students in Saudi Arabia. *Int. J. Gend. Entrep.* **2019**, *11*, 435–458. [CrossRef]
- Al-Kwifi, O.S.; Tien Khoa, T.; Ongsakul, V.; Ahmed, Z.U. Determinants of female entrepreneurship success across Saudi Arabia. J. Transnatl. Manag. 2020, 25, 3–29. [CrossRef]
- Basaffar, A.A.; Niehm, L.S.; Bosselman, R. Saudi Arabian women IN entrepreneurship: Challenges, opportunities and potential. J. Dev. Entrep. 2018, 23, 1850013. [CrossRef]
- Syed, A.M.; Alaraifi, A.; Ahmad, S. Entrepreneurs in Saudi Arabia: Risk attitude and predisposition towards risk management. J. Entrep. Educ. 2019, 22, 1–18.
- 8. Elnadi, M.; Gheith, M.H.; Farag, T. How does the perception of entrepreneurial ecosystem affect entrepreneurial intention among university students in Saudi Arabia? *Int. J. Entrep.* 2020, 24, 1–15.
- 9. Alwakid, W.; Aparicio, S.; Urbano, D. Cultural antecedents of green entrepreneurship in Saudi Arabia: An institutional approach. *Sustainability* **2020**, *12*, 3673. [CrossRef]
- 10. Armuña, C.; Ramos, S.; Juan, J.; Feijóo, C.; Arenal, A. From stand-up to start-up: Exploring entrepreneurship competences and STEM women's intention. *Int. Entrep. Manag. J.* **2020**, *16*, 69–92. [CrossRef]
- Mitchelmore, S.; Rowley, J. Entrepreneurial competencies: A literature review and development agenda. *Int. J. Entrep. Behav. Res.* 2010, 16, 92–111. [CrossRef]
- Kyndt, E.; Baert, H. Entrepreneurial competencies: Assessment and predictive value for entrepreneurship. J. Vocat. Behav. 2015, 90, 13–25. [CrossRef]
- 13. Man, T.W.; Lau, T.; Chan, K.F. The competitiveness of small and medium enterprises: A conceptualization with focus on entrepreneurial competencies. *J. Bus. Ventur.* 2002, *17*, 123–142. [CrossRef]
- Sánchez, J.C. The impact of an entrepreneurship education program on entrepreneurial competencies and intention. J. Small Bus. Manag. 2013, 51, 447–465. [CrossRef]
- Mitchelmore, S.; Rowley, J. Entrepreneurial competencies of women entrepreneurs pursuing business growth. J. Small Bus. Enterp. Dev. 2013, 20, 125–142. [CrossRef]
- 16. Man, T.W.; Lau, T.; Snape, E. Entrepreneurial competencies and the performance of small and medium enterprises: An investigation through a framework of competitiveness. *J. Small Bus. Entrep.* **2008**, *21*, 257–276. [CrossRef]
- 17. Plumly, L.W., Jr.; Marshall, L.L.; Eastman, J.; Iyer, R.; Stanley, K.L.; Boatwright, J. Developing entrepreneurial competencies: A student business. *J. Entrep. Educ.* 2008, *11*, 17–18.
- 18. Lingappa, A.K.; Rodrigues, L.R.L.; Shetty, D.K. Women entrepreneurial motivation and business performance: The role of learning motivation and female entrepreneurial competencies. *Ind. Commer. Train.* **2023**. [CrossRef]
- 19. Sánchez, J. The influence of entrepreneurial competencies on small firm performance. Rev. Latinoam. Psicol. 2012, 44, 165–177.
- Ahmad, N.H. A Cross Cultural Study of Entrepreneurial Competencies and Entrepreneurial Success in SMEs in Australia and Malaysia. Ph.D. Thesis, Adelaide Graduate School of Business, Adelaide, Australia, 2007. Available online: https://hdl.handle. net/2440/48199 (accessed on 21 December 2022).
- 21. Inyang, B.J.; Enuoh, R.O. Entrepreneurial competencies: The missing links to successful entrepreneurship in Nigeria. *Int. Bus. Res.* 2009, 2, 62–71. [CrossRef]
- Aftab, J.; Veneziani, M.; Sarwar, H.; Ishaq, M.I. Entrepreneurial orientation and firm performance in SMEs: The mediating role of entrepreneurial competencies and moderating role of environmental dynamism. *Int. J. Emerg. Mark.* 2022, *ahead-of-print*. [CrossRef]
- Al Koliby, M.I.S.; Abdullah, H.H.; Suki, M.N. Linking entrepreneurial competencies, innovation and sustainable performance of manufacturing SMEs. Asia-Pac. J. Bus. Adm. 2022, ahead-of-print. [CrossRef]
- Huynh, T.; Patton, D.; Arias-Aranda, D.; Molina-Fernández, L.M. University spin-off's performance: Capabilities and networks of founding teams at creation phase. J. Bus. Res. 2017, 78, 10–22. [CrossRef]
- Arias Aranda, D.; Bustinza Sanchez, O.F.; Djundubaev, R. Effects of gamified business simulations on entrepreneurial attitude at high school level. *Rev. Educ.* 2016, 371, 133–156.
- Rehman, S.U.; Elrehail, H.; Nair, K.; Bhatti, A.; Taamneh, A.M. MCS package and entrepreneurial competency influence on business performance: The moderating role of business strategy. *Eur. J. Manag. Bus. Econ.* 2021, 32, 1–123. [CrossRef]
- 27. Lim, W.; Lee, Y.; Mamun, A.A. Delineating competency and opportunity recognition in the entrepreneurial intention analysis framework. *J. Entrep. Emerg. Econ.* **2023**, *15*, 212–232. [CrossRef]
- 28. Haynie, J.M.; Shepherd, D.; Mosakowski, E.; Earley, P.C. A situated metacognitive model of the entrepreneurial mindset. *J. Bus. Ventur.* **2010**, 25, 217–229. [CrossRef]
- 29. Kuratko, D.F.; Fisher, G.; Audretsch, D.B. Unraveling the entrepreneurial mindset. Small Bus. Econ. 2021, 57, 1681–1691. [CrossRef]
- 30. Pihie ZA, L.; Sani AS, A. Exploring the entrepreneurial mindset of students: Implication for improvement of entrepreneurial learning at university. *J. Int. Soc. Res.* **2009**, *2*, 340–345.
- Zupan, B.; Cankar, F.; Setnikar Cankar, S. The development of an entrepreneurial mindset in primary education. *Eur. J. Educ.* 2018, 53, 427–439. [CrossRef]
- Pfeifer, S.; Šarlija, N.; Zekić Sušac, M. Shaping the entrepreneurial mindset: Entrepreneurial intentions of business students in Croatia. J. Small Bus. Manag. 2016, 54, 102–117. [CrossRef]

- 33. Larsen, I.B. Fostering an entrepreneurial mindset: A typology for aligning instructional strategies with three dominant entrepreneurial mindset conceptualizations. *Ind. High. Educ.* **2022**, *36*, 236–251. [CrossRef]
- Wardana, L.W.; Narmaditya, B.S.; Wibowo, A.; Mahendra, A.M.; Wibowo, N.A.; Harwida, G.; Rohman, A.N. The impact of entrepreneurship education and students' entrepreneurial mindset: The mediating role of attitude and self-efficacy. *Heliyon* 2020, 6, e04922. [CrossRef]
- 35. Hmieleski, K.M.; Lerner, D.A. The dark triad and nascent entrepreneurship: An examination of unproductive versus productive entrepreneurial motives. *J. Small Bus. Manag.* **2016**, *54*, 7–32. [CrossRef]
- Williams, C.C.; Round, J. Evaluating informal entrepreneurs' motives: Evidence from Moscow. Int. J. Entrep. Behav. Res. 2009, 15, 94–107. [CrossRef]
- 37. Kahraman, O.C.; Cifci, I.; Tiwari, S. Residents' entrepreneurship motives, attitude, and behavioral intention toward the mealsharing economy. *J. Hosp. Mark. Manag.* **2023**, *32*, 317–339. [CrossRef]
- Cachon, J.C.; Codina, J.B.; Eccius-Wellmann, C.; McGraw, E.; Myers, D.A. Entrepreneurial motives and performance: Evidence from North America. J. Manag. Policy Pract. 2013, 14, 50–77.
- Akinyemi, F.; Adejumo, O. Entrepreneurial motives and challenges of SMEs owners in emerging economies: Nigeria & South Africa. Small 2017, 5, 624–633.
- 40. Jayawarna, D.; Rouse, J.; Kitching, J. Entrepreneur motivations and life course. Int. Small Bus. J. 2013, 31, 34–56. [CrossRef]
- 41. Xu, Z.; Zhou, Y.; Zhang, Y.; Zhang, Y.; Ouyang, Z. Family–work enrichment and entrepreneurial intentions: A family affective support perspective. *Manag. Decis.* **2023**, *61*, 57–76. [CrossRef]
- 42. Ciptono, W.S.; Anggadwita, G.; Indarti, N. Examining prison entrepreneurship programs, self-efficacy and entrepreneurial resilience as drivers for prisoners' entrepreneurial intentions. *Int. J. Entrep. Behav. Res.* **2023**, *29*, 408–432. [CrossRef]
- Corvello, V.; Belas, J.; Giglio, C.; Iazzolino, G.; Troise, C. The impact of business owners' individual characteristics on patenting in the context of digital innovation. J. Bus. Res. 2023, 155, 113397. [CrossRef]
- Iazzolino, G.; Coniglio, I.M.; Verteramo, S.; Giglio, C. University students and entrepreneurship: An empirical analysis on Italian Universities. J. Entrep. Educ. 2019, 22, 1–16.
- 45. Ali, J.; Jabeen, Z.; Burhan, M. Measuring factors influencing entrepreneurial intention across gender in India: Evidence from Global Entrepreneurship Monitor (GEM) Database. *J. Res. Mark. Entrep.* **2023**, *25*, 63–82. [CrossRef]
- 46. Rehman, W.; Yosra, A.; Khattak, M.S.; Fatima, G. Antecedents and boundary conditions of entrepreneurial intentions: Perspective of theory of planned behaviour. *Asia Pac. J. Innov. Entrep.* **2023**. [CrossRef]
- Maes, J.; Leroy, H.; Sels, L. Gender differences in entrepreneurial intentions: A TPB multi-group analysis at factor and indicator level. *Eur. Manag. J.* 2014, 32, 784–794. [CrossRef]
- Soomro, B.A.; Shah, N. Developing attitudes and intentions among potential entrepreneurs. J. Enterp. Inf. Manag. 2015, 28, 304–322. [CrossRef]
- 49. Ali, B.; Shah, N.; Anwar, S. Investigating entrepreneurial intention through entrepreneurial event model (EEM) among graduate and master students in public sector universities of Thailand. *Asia Pac. Res. J. Far East South East Asia* 2016, *34*, 36–53.
- Soomro, B.A.; Lakhan, G.R.; Mangi, S.; Shah, N. Predicting entrepreneurial intention among business students of public sector universities of Pakistan: An application of the entrepreneurial event model. *World J. Entrep. Manag. Sustain. Dev.* 2020, 16, 219–230. [CrossRef]
- 51. Shahid, S. Perceived barriers and entrepreneurial exit intentions: Moderating role of regular versus sustainable entrepreneurship. *Eur. Bus. Rev.* **2023**, *35*, 39–56. [CrossRef]
- 52. Noor, U.; Rabbani, S.; Dastgeer, G. Impact of job insecurity during COVID-19 on green entrepreneurial intention of Pakistani entrepreneurs: A moderated mediation model. *Kybernetes* 2022, *ahead-of-print*. [CrossRef]
- 53. Dubey, P.; Sahu, K.K. Examining the effects of demographic, social and environmental factors on entrepreneurial intention. *Manag. Matters* **2022**, *19*, 91–108. [CrossRef]
- Mir, A.A.; Hassan, S.; Khan, S.J. Understanding digital entrepreneurial intentions: A capital theory perspective. *Int. J. Emerg. Mark.* 2022, *ahead-of-print*. [CrossRef]
- 55. Dubey, P. The effect of entrepreneurial characteristics on attitude and intention: An empirical study among technical undergraduates. J. Bus. Socio-Econ. Dev. 2022, ahead-of-print. [CrossRef]
- 56. Biswas, A.; Verma, R.K. Engine of entrepreneurial intentions: Revisiting personality traits with entrepreneurial education. *Benchmarking Int. J.* **2022**, *29*, 2019–2044. [CrossRef]
- Dragin, A.S.; Mijatov, M.B.; Munitlak Ivanović, O.; Jovičić Vuković, A.; Ivkov Džigurski, A.; Košić, K.; Stoiljković, A. Entrepreneurial intention of students (managers in training): Personal and family characteristics. *Sustainability* 2022, 14, 7345. [CrossRef]
- Ferri, L.; Ginesti, G.; Spanò, R.; Zampella, A. Exploring the entrepreneurial intention of female students in Italy. J. Open Innov. Technol. Mark. Complex. 2018, 4, 27. [CrossRef]
- Ahamed, F.; Rokhman, W. Personality and Entrepreneurial Motives of Islamic College Students: Environmental Support as Mediator. Int. J. Hum. Potentials Manag. 2019, 1, 1–15.
- 60. Nazmi, F.B.M. Self-entrepreneurial competencies and entrepreneurial mindset among university students. In Proceedings of the 7th International Conference on Management and Muamalah 2020 (ICoMM 2020), Bangi, Malaysia, 4 November 2020; pp. 86–129.

- 61. Cui, J.; Sun, J.; Bell, R. The impact of entrepreneurship education on the entrepreneurial mindset of college students in China: The mediating role of inspiration and the role of educational attributes. *Int. J. Manag. Educ.* **2021**, *19*, 100296. [CrossRef]
- 62. Kyrgidou, L.; Mylonas, N.; Petridou, E.; Vacharoglou, E. Entrepreneurs' competencies and networking as determinants of women-owned ventures success in post-economic crisis era in Greece. J. Res. Mark. Entrep. 2021, 23, 211–234. [CrossRef]
- Ananthram, S.; Pearson, C.; Chatterjee, S. Do organisational reform measures impact on global mindset intensity of managers? Empirical evidence from Indian and Chinese service industry managers. *J. Chin. Econ. Foreign Trade Stud.* 2010, *3*, 146–168. [CrossRef]
- 64. Bosman, L.; Kotla, B.; Cuesta, C.; Duhan, N.; Oladepo, T. The role of information literacy in promoting "discovery" to cultivate the entrepreneurial mindset. *J. Int. Educ. Bus.* **2023**, *16*, 56–69. [CrossRef]
- 65. Kakouris, A. Exploring entrepreneurial conceptions, beliefs and intentions of Greek graduates. *Int. J. Entrep. Behav. Res.* 2016, 22, 109–132. [CrossRef]
- Jabeen, F.; Faisal, M.N.; I. Katsioloudes, M. Entrepreneurial mindset and the role of universities as strategic drivers of entrepreneurship: Evidence from the United Arab Emirates. J. Small Bus. Enterp. Dev. 2017, 24, 136–157. [CrossRef]
- 67. Wafaretta, K.V.; Syariati, D. Enhancing cognitive and entrepreneurship competencies through EBL: Preparation for disruption era. *Asian J. Account. Res.* **2019**, *4*, 280–292.
- 68. Agyapong, A.; Maaledidong, P.D.; Mensah, H.K. Performance outcome of entrepreneurial behavior of SMEs in a developing economy: The role of international mindset. *J. Strategy Manag.* **2021**, *14*, 227–245. [CrossRef]
- 69. San-Martín, P.; Pérez, A.; Fernández-Laviada, A.; Palazuelos, E. The effect of role model teachers on students' entrepreneurial competencies and intentions. *Educ.* + *Train.* **2022**, *64*, 962–980. [CrossRef]
- 70. Asenge, E.L. Entrepreneurial competencies and entrepreneurial mindset as determinants of small and medium scale enterprises performance in Nigeria. *Glob. J. Manag. Bus. Res.* **2018**, *18*, 25–34.
- Sánchez, J.C. University training for entrepreneurial competencies: Its impact on intention of venture creation. *Int. Entrep. Manag.* J. 2011, 7, 239–254. [CrossRef]
- 72. Menke, C. Unravelling entrepreneurial competencies and their relation to entrepreneurial intent. *Int. J. Entrep. Ventur.* **2018**, *10*, 663–687. [CrossRef]
- Kusumawijaya, I.K.; Astuti, P.D. Mediating role of entrepreneurial competencies: Influence of personality traits on entrepreneurial intention. *Probl. Perspect. Manag.* 2021, 19, 211–220. [CrossRef]
- Rezaei, S. Entrepreneurial competencies benefiting entrepreneurial intention: Iranian adults at home and in the diaspora. *Iran.* Entrep. Deciphering Entrep. Ecosyst. Iran Iran. Diaspora 2017, ahead-of-print.
- 75. Carruthers, T.J.; Venter, M.W.; Botha, M. The relationship between entrepreneurial competencies and the recurring entrepreneurial intention and action of existing entrepreneurs. *S. Afr. J. Entrep. Small Bus. Manag.* **2019**, *11*, 1–15.
- Ismail, V.Y.; Zain, E. The portrait of entrepreneurial competence on student entrepreneurs. *Procedia-Soc. Behav. Sci.* 2015, 169, 178–188. [CrossRef]
- 77. Bastian, B.L.; Zali, M.R. Entrepreneurial motives and their antecedents of men and women in North Africa and the Middle East. *Gend. Manag.* **2016**, *31*, 456–478.
- Pulka, B.M.; Ramli, A.; Mohamad, A. Entrepreneurial competencies, entrepreneurial orientation, entrepreneurial network, government business support and SMEs performance. The moderating role of the external environment. *J. Small Bus. Enterp. Dev.* 2021, 28, 586–618. [CrossRef]
- 79. Ahmad, H.N.; Ramayah, T.; Wilson, C.; Kummerow, L. Is entrepreneurial competency and business success relationship contingent upon business environment? A study of Malaysian SMEs. *Int. J. Entrep. Behav. Res.* 2010, 16, 182–203. [CrossRef]
- Al Mamun, A.; Muniady, R.A.; Ibrahim, M.A.H.B.; Nawi, N.B.C. Effect of economic vulnerability on entrepreneurial competencies among Malaysian micro-entrepreneurs. Asia Pac. J. Innov. Entrep. 2018, 12, 222–237. [CrossRef]
- Priyanto, S.H.; Sandjojo, I. Relationship between entrepreneurial learning, entrepreneurial competencies and venture success: Empirical study on SMEs. Int. J. Entrep. Innov. Manag. 2005, 5, 454–468. [CrossRef]
- Krueger, N.F., Jr.; Reilly, M.D.; Carsrud, A.L. Competing models of entrepreneurial intentions. J. Bus. Ventur. 2000, 15, 411–432. [CrossRef]
- Tiwari, P.; Bhat, A.K.; Tikoria, J. Predictors of social entrepreneurial intention: An empirical study. South Asian J. Bus. Stud. 2017, 6, 53–79. [CrossRef]
- 84. Wickramaratne, A.; Kiminami, A.; Yagi, H. Entrepreneurial competencies and entrepreneurial orientation of tea manufacturing firms in Sri Lanka. *Asian Soc. Sci.* **2014**, *10*, 50–62. [CrossRef]
- 85. Solesvik, Z.M.; Westhead, P.; Matlay, H.; Parsyak, N.V. Entrepreneurial assets and mindsets: Benefit from university entrepreneurship education investment. *Educ.* + *Train.* **2013**, *55*, 748–762. [CrossRef]
- Adebusuyi, A.S.; Kolawole, S.O.; Abu, H.S.; Adebusuyi, O.F.; Ajulo, A.A. Predicting entrepreneurial and professional career mindsets in young Nigerian adults. J. Entrep. Emerg. Econ. 2022, 14, 1096–1117. [CrossRef]
- 87. Yodchai, N.; Ly, P.T.M.; Tran, L.T.T. How the creative mindset affects entrepreneurial success in the tourism sector: The mediating role of innovation capability. *Int. J. Contemp. Hosp. Manag.* 2022, 34, 279–298. [CrossRef]
- 88. Mukhtar, S.; Wardana, L.W.; Wibowo, A.; Narmaditya, B.S. Does entrepreneurship education and culture promote students' entrepreneurial intention? The mediating role of entrepreneurial mindset. *Cogent Educ.* **2021**, *8*, 1918849. [CrossRef]

- Cui, J.; Bell, R. Behavioural entrepreneurial mindset: How entrepreneurial education activity impacts entrepreneurial intention and behaviour. *Int. J. Manag. Educ.* 2022, 20, 100639. [CrossRef]
- 90. Wach, K.; Wojciechowski, L. Entrepreneurial intentions of students in Poland in the view of Ajzen's theory of planned behaviour. *Entrep. Bus. Econ. Rev.* **2016**, *4*, 83–94. [CrossRef]
- Munawar, S.; Yousaf, H.Q.; Ahmed, M.; Rehman, S. The influence of online entrepreneurial education on entrepreneurial success: An empirical study in Pakistan. Int. J. Manag. Educ. 2023, 21, 100752. [CrossRef]
- Asghar, M.Z.; Gul, F.; Seitamaa-Hakkarainen, P.; Tasdemir, M.Z. Validating entrepreneurial intentions questionnaire to assess the impact of entrepreneurship education. *Eğitim Bilim* 2019, 44, 383–399. [CrossRef]
- 93. Shi, Y.; Yuan, T.; Bell, R.; Wang, J. Investigating the relationship between creativity and entrepreneurial intention: The moderating role of creativity in the theory of planned behavior. *Front. Psychol.* **2020**, *11*, 1209. [CrossRef] [PubMed]
- 94. Salami, S.O. Examining the emerging entrepreneurial mindset in adolescence: A study in Nigeria. *Int. J. Psychol.* **2019**, *54*, 70–79. [CrossRef]
- 95. Ruskin, J.; Seymour, R.G.; Webster, C.M. Why create value for others? An exploration of social entrepreneurial motives. *J. Small Bus. Manag.* 2016, 54, 1015–1037. [CrossRef]
- Chen, S.C.; Elston, J.A. Entrepreneurial motives and characteristics: An analysis of small restaurant owners. *Int. J. Hosp. Manag.* 2013, 35, 294–305. [CrossRef]
- 97. Şeşen, H.; Pruett, M. The impact of education, economy and culture on entrepreneurial motives, barriers and intentions: A comparative study of the United States and Turkey. *J. Entrep.* **2014**, *23*, 231–261. [CrossRef]
- Aziz, N.; Friedman, B.A.; Bopieva, A.; Keles, I. Entrepreneurial motives and perceived problems: An empirical study of entrepreneurs in Kyrgyzstan. *Int. J. Bus.* 2013, 18, 163–176.
- Saptono, A.; Wibowo, A.; Narmaditya, B.S.; Karyaningsih RP, D.; Yanto, H. Does entrepreneurial education matter for Indonesian students' entrepreneurial preparation: The mediating role of entrepreneurial mindset and knowledge. *Cogent Educ.* 2020, 7, 1836728. [CrossRef]
- 100. Saadat, S.; Aliakbari, A.; Alizadeh Majd, A.; Bell, R. The effect of entrepreneurship education on graduate students' entrepreneurial alertness and the mediating role of entrepreneurial mindset. *Educ.* + *Train.* **2022**, *64*, 892–909. [CrossRef]
- Ngek, N.B. Entrepreneurial self-efficacy (ESE) and small business performance: The mediating effect of entrepreneurial mindset and openness to experience. *Probl. Perspect. Manag.* 2015, 13, 271–280.
- 102. Altahat, S.; Alsafadi, Y. The mediating role of creativity in the influence of the entrepreneurial mindset on corporate entrepreneurship. *Int. J. Innov. Creat. Chang.* **2021**, *15*, 767–783.
- 103. Handayati, P.; Wulandari, D.; Soetjipto, B.E.; Wibowo, A.; Narmaditya, B.S. Does entrepreneurship education promote vocational students' entrepreneurial mindset? *Heliyon* 2020, *6*, e05426. [CrossRef] [PubMed]
- Marjerison, R.K.; Chen, R.; Lin, Y. The nexus of social cause interest and entrepreneurial mindset: Driving socioeconomic sustainability. *Sustainability* 2021, 13, 13558. [CrossRef]
- 105. Hassan, A.; Anwar, I.; Saleem, I.; Islam, K.B.; Hussain, S.A. Individual entrepreneurial orientation, entrepreneurship education and entrepreneurial intention: The mediating role of entrepreneurial motivations. *Ind. High. Educ.* 2021, 35, 403–418. [CrossRef]
- 106. Lee, Y.J.; Hwang, H. Keep the fire burning: Social entrepreneurs' motives, role stress, perceived work competence and passion. J. Gen. Manag. 2022. [CrossRef]
- 107. Bryman, A.; Bell, E. Business Research Methods; Oxford University Press Inc.: New York, NY, USA, 2007.
- 108. Denzin, N.K.; Lincoln, Y.S. (Eds.) Handbook of Qualitative Research; Sage: Thousand Oaks, CA, USA, 1994.
- 109. Kusumajanto, D.D. The role of entrepreneurship education and self-efficacy in shaping student interest in doing entrepreneurship. *Eur. J. Bus. Manag.* **2015**, *7*, 153–158.
- 110. Hoang, G.; Le, T.T.T.; Tran, A.K.T.; Du, T. Entrepreneurship education and entrepreneurial intentions of university students in Vietnam: The mediating roles of self-efficacy and learning orientation. *Educ.* + *Train.* **2020**, *63*, 115–133. [CrossRef]
- Shah, N.; Soomro, B.A. Investigating entrepreneurial intention among public sector university students of Pakistan. *Educ.* + *Train.* 2017, 59, 841–855. [CrossRef]
- 112. Saiqal NY Al Ryan, J.C.; Parcero, O.J. Entrepreneurial intention and UAE youth: Unique influencers of entrepreneurial intentions in an emerging country context. *J. East-West Bus.* **2019**, *25*, 144–165. [CrossRef]
- 113. Elnadi, M.; Gheith, M.H. Entrepreneurial ecosystem, entrepreneurial self-efficacy, and entrepreneurial intention in higher education: Evidence from Saudi Arabia. *Int. J. Manag. Educ.* **2021**, *19*, 100458. [CrossRef]
- Bakar AR, A.; Ahmad, S.Z.; Wright, N.S.; Skoko, H. The propensity to business startup: Evidence from global entrepreneurship monitor (GEM) data in Saudi Arabia. J. Entrep. Emerg. Econ. 2017, 9, 263–285. [CrossRef]
- 115. House, F. Women's Rights in the Middle East and North Africa: Citizenship and Justice; Rowman & Littlefield Publishers: Washington, DC, USA, 2005.
- Choukir, J.; Aloulou, W.J.; Ayadi, F.; Mseddi, S. Influences of role models and gender on Saudi Arabian freshman students' entrepreneurial intention. *Int. J. Gend. Entrep.* 2019, 11, 186–206. [CrossRef]
- Akinwale, Y.O.; Ababtain, A.K.; Alaraifi, A.A. Structural equation model analysis of factors influencing entrepreneurial interest among university students in Saudi Arabia. J. Entrep. Educ. 2019, 22, 1–14.
- Ahmed, S.M. Artificial intelligence in Saudi Arabia: Leveraging entrepreneurship in the Arab markets. In Proceedings of the 2019 Amity International Conference on Artificial Intelligence (AICAI), Dubai, United Arab Emirates, 4–6 February 2019; pp. 394–398.

- 119. Saunders, A.; Swinerd, G.G.; Lewis, H.G. Deriving accurate satellite ballistic coefficients from two-line element data. *J. Spacecr. Rocket.* 2012, 49, 175–184. [CrossRef]
- 120. Henson, R.K. Understanding internal consistency reliability estimates: A conceptual primer on coefficient alpha. *Meas. Eval. Couns. Dev.* **2001**, *34*, 177–189. [CrossRef]
- 121. Hair, J.F.; Risher, J.J.; Sarstedt, M.; Ringle, C.M. When to use and how to report the results of PLS-SEM. *Eur. Bus. Rev.* 2019, *31*, 2–24. [CrossRef]
- 122. Oliver, V. 301 Smart Answers to Tough Business Etiquette Questions; Skyhorse Publishing: New York, NY, USA, 2010.
- 123. Robson, C. Real World Research: A Resource for Users of Social Research Methods in Applied Settings, 2nd ed.; Sussex, A., Ed.; John Wiley and Sons Ltd.: Hoboken, NJ, USA, 2011.
- 124. Rubio, D.M.; Berg-Weger, M.; Tebb, S.S.; Lee, E.S.; Rauch, S. Objectifying content validity: Conducting a content validity study in social work research. *Soc. Work Res.* 2003, 27, 94–104. [CrossRef]
- 125. Medina-Díaz, M.D.R.; Verdejo-Carrión, A.L. Validity and reliability in student learning evaluation throughout active methodologies. *ALTERIDAD Rev. Educ.* 2020, 15, 270–284. [CrossRef]
- 126. Arora, N.; Aggarwal, A. The role of perceived benefits in formation of online shopping attitude among women shoppers in India. *South Asian J. Bus. Stud.* **2018**, *7*, 91–110. [CrossRef]
- 127. Cunningham, J.B.; McCrum-Gardner, E. Power, effect and sample size using GPower: Practical issues for researchers and members of research ethics committees. *Evid. Based Midwifery* **2007**, *5*, 132–136.
- 128. Erdfelder, E.; Faul, F.; Buchner, A. GPOWER: A general power analysis program. *Behav. Res. Methods Instrum. Comput.* **1996**, *28*, 1–11. [CrossRef]
- 129. Hair Joseph, F., Jr.; Black William, C.; Babin Barry, J.; Anderson Rolph, E.; Tatham Ronald, L. *Multivariate Data Analysis*, 6th ed.; Pearson Prentice Hall, Pearson Education, Inc.: Upper Saddle River, NJ, USA, 2006.
- 130. Elam, A.B. *Women's Entrepreneurship 2020/21: Thriving through Crisis;* Global Entrepreneurship Research Association: London, UK, 2021.
- López-Núñez, M.I.; Rubio-Valdehita, S.; Armuña, C.; Pérez-Urria, E. EntreComp questionnaire: A self-assessment tool for entrepreneurship competencies. *Sustainability* 2022, 14, 2983. [CrossRef]
- 132. Farida, F.A.; Hermanto, Y.B.; Paulus, A.L.; Leylasari, H.T. Strategic entrepreneurship mindset, strategic entrepreneurship leadership, and entrepreneurial value creation of SMEs in east Java, Indonesia: A strategic entrepreneurship perspective. *Sustainability* **2022**, *14*, 10321. [CrossRef]
- 133. Daspit, J.J.; Fox, C.J.; Findley, S.K. Entrepreneurial mindset: An integrated definition, a review of current insights, and directions for future research. *J. Small Bus. Manag.* **2021**, *59*, 1–33. [CrossRef]
- 134. Walter, S.G.; Dohse, D. Why mode and regional context matter for entrepreneurship education. *Entrep. Reg. Dev.* **2012**, *24*, 807–835. [CrossRef]
- 135. Volery, T.; Müller, S.; Oser, F.; Naepflin, C.; Rey, N.D. The impact of entrepreneurship education on human capital at uppersecondary level. *J. Small Bus. Manag.* 2013, *51*, 429–446. [CrossRef]
- Berns, J. Killing Two Birds with One Stone. Exploring Refugee Entrepreneurial Intent in The Netherlands. Master's Thesis, Radboud University, Nijmegen, The Netherlands, 2017. (Unpublished).
- 137. Boldureanu, G.; Lache, C.; Boldureanu, D.; Păduraru, T.; Niculescu, N. Students' entrepreneurial competencies and orientation: Current status and perspectives. *Environ. Eng. Manag. J.* **2013**, *12*, 2071–2077. [CrossRef]
- 138. Khalifa, A.H.; Dhiaf, M.M. The impact of entrepreneurship education on entrepreneurial intention: The UAE context. *Pol. J. Manag. Stud.* **2016**, *14*, 119–128. [CrossRef]
- 139. Alshebami, A.S. Psychological features and entrepreneurial intention among Saudi small entrepreneurs during adverse times. *Sustainability* **2022**, *14*, 7604. [CrossRef]
- 140. Sharma, S.; Mukherjee, S.; Kumar, A.; Dillon, W.R. A simulation study to investigate the use of cutoff values for assessing model fit in covariance structure models. *J. Bus. Res.* 2005, *58*, 935–943. [CrossRef]
- 141. Tomarken, A.J.; Waller, N.G. Potential problems with "Well Fitting" Models. J. Abnorm. Psychol. 2003, 112, 578–598. [CrossRef] [PubMed]
- 142. Yuan, K.H. Fit indices versus test statistics. Multivar. Behav. Res. 2005, 40, 115–148. [CrossRef] [PubMed]

**Disclaimer/Publisher's Note:** The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.